Author's response to reviews

Title: Left atrioventricular remodeling in the assessment of the left ventricle diastolic function in patients with heart failure: a review of the currently studied echocardiographic variables.

Authors:

Luiz C Danzmann (danzmann@redemeta.com.br)
Luiz Carlos Bodanese (lcbodanese@pucrs.br)
Ilmar Köhler (i.kohler@terra.com.br)
Marco R Torres (marcoartorres@yahoo.com)

Version: 6 Date: 5 November 2008

Author's response to reviews:

Dear Mr Editor

The authors revised the manuscript and they did 3 small corrections:
1) The abbreviation of the author Bodanese was corrected (it was LCD and it was changed to LCB);
2) An acknowledgement paragraph was inserted at the final of the manuscript
3) References were corrected with removal with reference number 2 - as previously suggested by the reviewers, change of the reference of Mitchell by the original reference of Zile (as it was at the first versions).

Now everything else looks well.

Sincerely yours

Dr Marco Torres